Author’s response to reviews

Title: Pelvic radiotherapy for cervical cancer affects importantly the reproducibility of cytological alterations evaluation.

Authors:
Fernanda Lucena (feraireslucena@gmail.com)
Ricardo Filipe Alves Costa (ricardofacosta@gmail.com)
Maira Degiovani Stein (mairastein@hotmail.com)
Carlos Eduardo Mattos Cunha Andrade (mdcarlosandrade@gmail.com)
Geórgia Fontes Cintra (georgia.fc@gmail.com)
Marcelo Andrade Vieira (mvieiraonco@gmail.com)
Rozany Mucha Duflotoh (rozany.dufloth@gmail.com)
Jose Humberto Tavares Guerreiro Fregnani (mdfregnani@terra.com.br)
Ricardo dos Reis (drricardoreis@gmail.com)

Version: 3 Date: 06 Sep 2018

Author’s response to reviews:

1. Please include the email addresses of all authors on the title page of your manuscript. The corresponding author should still be indicated. 
Answer = All email addresses was included.

2. Please amend the heading Introduction to Background. 
Answer: This amendment was corrected at the text.

3. Please amend the heading Materials and Methods to Methods 
Answer: This amendment was corrected at the text.

4. Thank you for indicating that consent was not needed in the Methods. Please also place this statement in the Ethics approval and consent to participate section. 
Answer: This amendment was included at the text.

5. Please provide more justification for the contributions of authors. Currently, the contributions of
MA, CA, and GF do not automatically qualify for authorship. Please amend your manuscript accordingly.
Answer: The contributions of MA, CA and GF was corrected as they participated in the conception and design of the study. Also, CA was involved in drafting the manuscript and revising it critically for important intellectual content.

6. We notice you have listed 8 initials along with the statement that authors read and approved the final version, but have 9 authors. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.
Answer: All authors read and given approval to the final version. This statement was corrected at the text.

7. Please remove any figure titles and captions from the figure files as these files should contain the image graphics only. Upon doing this, please place the titles/captions at the end of the main manuscript after the References.
Answer: The figure file was correct as required above. Also, the titles was included at the end of the main manuscript after references section.

8. Please remove the point by point letter as this is no longer needed at this stage in the editorial process.
Answer: This documents was removed as required.

9. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Answer: The manuscript will be upload as required above, also with Tables, Supplementary Tables and Figure files.

Editor Comments:

The authors have satisfactorily addressed the reviewers's comments.

Answer : On behalf of my co-authors, Fernanda Aires Lucena is pleased to receive the information that our revised version of manuscript entitled “Pelvic radiotherapy for cervical cancer affects importantly the reproducibility of cytological alterations evaluation.” is potentially acceptable for publication in BMC Clinical Pathology. We have carried your essential revisions and corrected details as suggested. Thank you very much for another reconsideration.